Author's response to reviews

Title: Cardiac involvement in beagle-based canine X-linked muscular dystrophy in Japan (CXMDJ): electrocardiographic, echocardiographic, and morphologic studies

Authors:

Naoko Yugeta (yugeta@b-star.jp)
Nobuyuki Urasawa (urasawan@amber.plala.or.jp)
Yoko Fujii (fujii@dc4.so-net.ne.jp)
Madoka Yoshimura (yoshimur@ncnp.go.jp)
Katsutoshi Yuasa (yuasa@ncnp.go.jp)
Akinori Nakamura (anakamu@ncnp.go.jp)
Michiko Wada (wadarin@ncnp.go.jp)
Masao Nakura (nakuramso@chugai-pharm.co.jp)
Yoshiki Shimatsu (shimatsu@ncnp.go.jp)
Masayuki Tomohiro (tomhromy@banyu.co.jp)
Akio Takahashi (akio_tk@ncnp.go.jp)
Noboru Machida (machida@cc.tuat.ac.jp)
Yoshito Wakao (wakao@azabu-u.ac.jp)
Shin'ichi Takeda (takeda@ncnp.go.jp)

Version: 2 Date: 27 July 2006

Author's response to reviews: see over
Dear Editor,

Thank you very much for your suggestion in the e-mail on July 20, 2006. I revised and uploaded our manuscript (MS: 1681285926109043), which is containing a ‘Competing interests’ section and an ‘Authors’ contributions’, based on your comments. We also removed the 'post-it' like notes.

I appreciate your review of this work.

Sincerely yours,

Shin’ichi Takeda, M.D., Ph.D.

Department of Molecular Therapy,
National Institute of Neuroscience,
National Center of Neurology and Psychiatry,
4-1-1 Ogawahigashi-cho, Kodaira,
Tokyo 187-8502, Japan.
Tel: +81-42-346-1720
Fax: +81-42-346-1750
E-mail: takeda@ncnp.go.jp